Author's response to reviews

Title: Maternal hyperuricemia in normotensive singleton pregnancy, a prenatal finding with continuous perinatal and postnatal effects, a prospective cohort study

Authors:

Elaheh Amini (E_amini@yahoo.com)
Mahdi Sheikh (mahdisheikh@gmail.com)
Sedigheh Hantoushzadeh (hantoushzadeh@tums.ac.ir)
Mamak Shariat (mshariat@tums.ac.ir)
Alireza Abdollahi (a_abdollahi@tums.ac.ir)
Maryam Kashanian (maryamkashanian@yahoo.com)

Version: 3
Date: 25 February 2014

Author's response to reviews: see over
Dear Sir/Madam

Thank you very much for your kind consideration of our paper entitled "Maternal hyperuricemia in normotensive singleton pregnancy, a prenatal finding with continues perinatal and postnatal effects, a prospective cohort study" with the manuscript number of "MS: 4511397921179460". Here I am submitting the revised version of the paper that was changed according to your recommendations. The following change was made to the paper:

1. The previous title was grammatically incorrect, so the title is changed to the follows: "Maternal hyperuricemia in normotensive singleton pregnancy, a prenatal finding with continuous perinatal and postnatal effects, a prospective cohort study"

Thank you again for your kind consideration

Yours Sincerely

Mahdi Sheikh, M.D.

Corresponding Author

Maternal, fetal and neonatal Research Center,

Tehran University of Medical Sciences,

Tehran, Iran